Author's response to reviews

Title: Identifying binge drinkers based on parenting dimensions and alcohol-specific parenting practices: building classifiers on adolescent-parent paired data

Authors:

Rik Crutzen (Rik.Crutzen@maastrichtuniversity.nl)
Philippe J Giabbanelli (philippe.giabbanelli@mrc-epid.cam.ac.uk)
Astrid Jander (astrid.jander@maastrichtuniversity.nl)
Liesbeth Mercken (liesbeth.mercken@maastrichtuniversity.nl)
Hein de Vries (hein.devries@maastrichtuniversity.nl)

Version: 2
Date: 29 January 2015

Author's response to reviews:

Dear Proel Hennessy Vargas,

We have now uploaded a revised version containing the requested changes:
1. We have included line and page numbers.
2. We have moved figures and tables after the reference section.

On behalf of all authors,

Sincerely,

Rik Crutzen
Maastricht University/CAPHRI